Author’s response to reviews

Title: Is hospital discharge administrative data an appropriate source of information for cancer registries purposes? Some insights from four Spanish registries.

Authors:

- Enrique Bernal-Delgado (ebernal.iacs@aragon.es)
- Carmen Martos (cmartos@aragon.es)
- Natalia Martínez (nmartinez.iacs@aragon.es)
- María Dolores Chirlaque (MDolores.Chirlaque@carm.es)
- Mirari Márquez (mirari.marquez@carm.es)
- Carmen Navarro (Carmen.Navarro@carm.es)
- Lauro Hernando (lauro.hernando@carm.es)
- Joaquín Palomar (joaquin.palomar@carm.es)
- Isabel Izarzugaza (info5-san@ej-gv.es)
- Nerea Larrañaga (epidem3-san@ej-gv.es)
- Olatz Mokoroa (o-mokoroa@ej-gv.es)
- M Cres Tobalina (evalua3vi-san@ej-gv.es)
- Joseba Bidaurrazaga (josebabidaurrazaga@ej-gv.es)
- María José Sánchez (mariajose.sanchez.easp@juntadeandalucia.es)
- Carmen Martínez (carmen.martinez.easp@juntadeandalucia.es)
- Miguel Rodríguez (miguel.rodriguez.barranco.easp@juntadeandalucia.es)
- Esther Pérez (esther.perez.easp@juntadeandalucia.es)
- Yoe Ling Chang (daisy.chang.easp@juntadeandalucia.es)

Version: 6 Date: 11 December 2009

Author’s response to reviews:

Dear editor,

A new version of the manuscript has been uploaded. Major and minor revisions have been implemented. Some typographical mistakes (double line space, unjustifying texts, headlines in proper case editing, etc.) have also been corrected.

It was impossible to find an functioning link for reference #26. It was taken away.

Many thanks

Enrique Bernal-Delgado